

Strong Intensification of the Arabian Sea Oxygen Minimum Zone in Response to Arabian Gulf Warming

Zouhair Lachkar, Marina Lévy, K. Shafer Smith

▶ To cite this version:

Zouhair Lachkar, Marina Lévy, K. Shafer Smith. Strong Intensification of the Arabian Sea Oxygen Minimum Zone in Response to Arabian Gulf Warming. Geophysical Research Letters, 2019, 46, pp.5420 - 5429. 10.1029/2018GL081631 . hal-02192445

HAL Id: hal-02192445 https://hal.science/hal-02192445v1

Submitted on 1 Dec 2020

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Geophysical Research Letters



RESEARCH LETTER

10.1029/2018GL081631

Key Points:

- Highly saline, oxygen-saturated Arabian Gulf waters ventilate the Arabian Sea OMZ
- Warming of the Gulf causes reduction of ventilation and expansion of suboxia
- We find nonlinear response of suboxia to warming due to denitrification feedback

Supporting Information:

Supporting Information S1

Correspondence to:

Z. Lachkar, zouhair.lachkar@nyu.edu

Citation:

Lachkar, Z., Lévy, M., & Smith, K. S. (2019). Strong intensification of the Arabian Sea oxygen minimum zone in response to Arabian Gulf warming. *Geophysical Research Letters*, 46, 5420–5429. https://doi.org/10.1029/ 2018GL081631

Received 9 DEC 2018 Accepted 6 MAY 2019 Accepted article online 8 MAY 2019 Published online 29 MAY 2019

©2019. The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

Strong Intensification of the Arabian Sea Oxygen Minimum Zone in Response to Arabian Gulf Warming

Z. Lachkar¹, M. Lévy², and K. S. Smith^{1,3}

¹Center for Prototype Climate Modeling, New York University in Abu Dhabi, Abu Dhabi, UAE, ²Sorbonne Université (CNRS/IRD/MNHN), LOCEAN-IPSL, Paris, France, ³Courant Institute of Mathematical Sciences, New York University, New York, NY, USA

Abstract The highly saline, oxygen-saturated waters of the Arabian Gulf (hereafter the Gulf) sink to intermediate depths (200–300 m) when they enter the Arabian Sea, ventilating the World's thickest oxygen minimum zone (OMZ). Here, we investigate the impacts of a warming of the Gulf consistent with climate change projections on the intensity of this OMZ. Using a series of eddy-resolving model simulations, we show that the warming of the Gulf waters increases their buoyancy and hence limits their contribution to the ventilation of intermediate depths. This leads to an intensification of the OMZ and an increase in denitrification that depletes subsurface nitrate and limits deoxygenation at depth. The projected future concomitant increase of Gulf salinity only partially reduces the OMZ intensification. Our findings highlight the importance of the Arabian marginal seas for the biogeochemistry of the North Indian Ocean and stress the need for improving their representation in global climate models.

Plain Language Summary Dissolved oxygen in the ocean is fundamental for marine life. While relatively abundant in surface waters, oxygen generally declines with depth as it is consumed by organisms' respiration. In certain regions like the Arabian Sea, oxygen concentrations are too low at depth to support marine animals. These are known as "oxygen minimum zones" (OMZs). At their core, extreme oxygen depletion known as suboxia can also cause a loss of bioavailable nitrogen, essential for phytoplankton growth. Using a series of computer simulations, we show that the sinking of oxygen-saturated dense waters formed in the Arabian Gulf contributes to oxygen replenishment of the intermediate depths (200–300 m) in the northern Arabian Sea, reducing the intensity of the OMZ and limiting the volume of its suboxic core. We also show that a warming of the Gulf waters consistent with recent observations and future climate projections limits their ability to sink and ventilate the intermediate depths. This results in a strong intensification of the OMZ and an important loss of bioavailable nitrogen. Our findings highlight the importance of semienclosed seas like the Arabian Gulf for the ventilation of the ocean and hence stress the need for improving their representation in climate models.

1. Introduction

The Arabian Sea hosts the World's thickest oxygen minimum zone (OMZ), with oxygen concentrations being in the suboxic range (<4 mmol/m³) throughout most of the intermediate ocean (from 150 down to 1,250 m). This widespread suboxia is due to very high rates of biological respiration in the water column combined with a relatively poor intermediate water ventilation. Although it occupies less than 2% of the World Ocean area, the Arabian Sea is responsible for up to 40% of global water column denitrification, a process that depletes the oceanic inventory of bioavailable nitrogen (essential for phytoplankton growth) and releases N₂O (a major greenhouse gas; Bange et al., 2005; Codispoti et al., 2001). The main source of ventilation for intermediate waters in the northern Arabian Sea is the Indian Ocean Central Water—a mixture of aged Antarctic Intermediate Water, Subantarctic Mode Water, and Indonesian Intermediate Water—advected from the south by the Somali current (Olson et al., 1993; Resplandy et al., 2012; You, 1998). Another source of Arabian Sea intermediate water ventilation is the dense and highly saline waters from the Red Sea and the Arabian Gulf, also known as the Persian Gulf (hereinafter referred to as the Gulf; Bower et al., 2000, 2005; Prasad et al., 2001). The denser Red Sea waters ($\sigma_0 \approx 27.2$) sink to depths of 600–1,000 m and mix extensively with other water masses of the Arabian Sea before reaching the OMZ. In contrast, the Gulf waters $(\sigma_0 \approx 26.4)$ remain at shallower depths (150–300 m) and ventilate the upper OMZ more efficiently due to the closer proximity of the Gulf to the Arabian Sea OMZ core (Beal et al., 2000; Morrison et al., 1999; McCreary et al., 2013).

As a semienclosed sea with a very shallow depth (i.e., 35 m on average), the Gulf is particularly vulnerable to amplified warming under climate change. Indeed, recent observations indicate a rate of warming of the Gulf that is 2 to 3 times faster than the average global rate. For instance, Al Rashidi et al. (2009) documented a recent warming of the Bay of Kuwait between 1985 and 2002 at a rate of 0.6 °C per decade on the basis of an analysis of local in situ temperature records. An analysis of sea surface temperature (SST) from the National Oceanic and Atmospheric Administration/National Aeronautics and Space Administration Pathfinder product reveals an average warming of the Gulf of 0.6 °C per decade between 1985 and 2009 (Strong et al., 2011). Furthermore, future climate projections suggest that the Gulf may undergo a substantial warming that exceeds 3 °C and 4 °C by the end of the century under moderate mitigation emission scenario, respectively (Hoegh-Guldberg et al., 2014).

Recent observations suggest that the Arabian Sea OMZ may have intensified in the Gulf of Oman over the recent decades. For instance, Ito et al. (2017) show a moderate intensification of the Arabian Sea OMZ over the period from 1960 to 2010. Gomes et al. (2014) documented a dramatic shift in the winter bloom dominant species in the northern Arabian Sea from diatoms to large, green dinoflagellate, Noctiluca Scintillans, which combines photosynthesis from its endosymbiont with ingestion of prey. They linked this change to dropping oxygen concentrations in the region over the same period that give Noctiluca endosymbiont a competitive advantage over diatoms. Another observational study by Piontkovski and Al-Oufi (2015) also documented a recent decline of oxygen and a shoaling of the hypoxic boundary in the Sea of Oman. Finally, using a sea glider survey and historical data, Queste et al. (2018) also suggested a recent intensification of the OMZ in the Gulf of Oman over the last three decades.

The intensification of the Arabian Sea OMZ may not only affect the ecosystem and compress marine habitats but could also lead to enhanced denitrification, thus potentially altering the large-scale marine nitrogen budget and carbon cycle. Yet, the drivers of O_2 decline in this region are still poorly known. In particular, the contribution of the Gulf to the ventilation of the Arabian Sea OMZ remains poorly understood. Here we explore the importance of the Gulf in supplying O_2 to the Arabian Sea OMZ and examine the potential role of the Gulf warming and how it may contribute to these changes. We show that a moderate-to-strong warming of the Gulf, which is consistent with future emission scenarios, can lead to a substantial intensification of the Arabian Sea OMZ and of denitrification. These changes can have important ecological and biogeochemical consequences.

2. Methods

We use version 3.1.1 of the Regional Ocean Modeling System-AGRIF (more details available at http:// www.romsagrif.org/). The Regional Ocean Modeling System is a free-surface primitive equations model with orthogonal curvilinear coordinates in the horizontal and stretched terrain following coordinates in the vertical (Shchepetkin & McWilliams, 2005). Advection is formulated using a rotated-split third-order upstream biased operator following Marchesiello et al. (2009). Vertical mixing is represented using the nonlocal K-Profile Parameterization scheme (Large et al., 1994). The biogeochemical model is a nutrient-phytoplankton-zooplankton-detritus model based on nitrogen (Gruber et al., 2006). It includes an oxygen cycle module that is described in detail in Lachkar et al. (2016). At suboxic O₂ concentrations (O₂ < 4 mmol/m³) nitrification ceases and the remineralization rate is halved. Additionally, aerobic remineralization of detritus is replaced by water column denitrification, where nitrate replaces oxygen as the electron acceptor. Finally, benthic denitrification is represented in the model following the parameterization of Middelburg et al. (1996). More details of the implementation of denitrification in the model are given in Lachkar et al. (2016).

The model domain covers the area extending from $5^{\circ}S$ to $30^{\circ}N$ in latitude and from $32^{\circ}E$ to $78^{\circ}W$ in longitude. The model has a horizontal resolution of $(1/24)^{\circ}$ and a vertical grid made of 32 levels with refined resolution near the surface. The seafloor bathymetry is based on the ETOPO2 data product from the National Geophysical Data Center (Smith & Sandwell, 1997). Smoothing was applied to the topography to minimize pressure gradient errors near steep bathymetry. The model was initialized from rest and forced with



Figure 1. (a–d) Simulated O₂ concentrations (mmol/m³) in (left) the Control run and (right) the AG^{Dry-Gulf} run along (top) the isopycnal surface $\sigma_0 = 26.4$ and (bottom) a meridional transect at 60.2°E. The boundaries of the suboxic zone (i.e., O₂ = 4 mmol/m³) simulated under different warming scenarios are superimposed and shown in different colors in (a) and (c). The white contour lines in (c) and (d) correspond to potential density σ_0 surfaces. (e) Relative change (%) of domain-integrated suboxic volume in the top 200 m (solid) and below 200 m (dashed) under different perturbation simulations relative to the control run. (f) Potential density σ_0 (red) and salinity (blue) simulated in the AG^{Dry-Gulf} plotted against the control run in the layer 100–1,000 m. The dashed black line indicates the identity line.

climatological monthly forcing. Wind stress was derived from the QuikSCAT-derived Scatterometer Climatology of Ocean Winds (Risien & Chelton, 2008). The surface heat and freshwater fluxes are based on the Comprehensive Ocean-Atmosphere Data Set (COADS; da Silva et al., 1994). Surface temperature and salinity were restored to COADS observations using a kinematic heat and freshwater flux corrections following Barnier et al. (1995). The lateral boundary conditions for currents and the initial and boundary conditions for temperature and salinity were derived from the Simple Ocean Data Assimilation ocean reanalysis. The initial and lateral boundary conditions for oxygen and nitrate are based on the World Ocean Atlas 2009.



Figure 2. (a–e) Simulated potential density σ_0 across the Strait of Hormuz (56.2°E) in (a) the control run and (b–e) under different warming scenarios. The black contour lines show the corresponding O₂ concentrations (mmol/m³). (f) Simulated depth of neutrally buoyant Gulf waters (exiting the Strait of Hormuz below 60 m) along 60°E in the control run and under different warming scenarios.

The model is first spun up for 20 years and then is run for an additional 20 years using different perturbation scenarios. In addition to the control run (Control), a first perturbation simulation that consists in closing the Gulf by modifying its bathymetry (AG^{Dry-Gulf}) was performed. The comparison of this first set of simulations aims at evaluating the overall contribution of the Gulf to the ventilation of the Arabian Sea OMZ. In a second set of warming perturbation simulations, the topography was set identical to the control run but the Gulf SST was restored to COADS observations increased by 1 °C (AG^{1C}), 2 °C (AG^{2C}) and 4°C (AG^{4C}), respectively. An additional run was conducted where SST was forced to increase by 2 °C over the Gulf and $1 \,^{\circ}$ C over the rest of the domain (AG^{2C*}). This is aimed at exploring the effect of a weaker concurrent surface temperature increase in the Arabian Sea on the response of the OMZ to the Gulf warming. These runs represent highly idealized scenarios that are not meant to describe realistic future trajectories but rather aim at exploring the sensitivity of the Arabian Sea OMZ to Gulf water temperature changes and improving our understanding of the key mechanisms that control the OMZ response. For model evaluation and oxygen budgets, we generally use the model years 33 to 40. The control run reproduces fairly accurately the observed spatial and temporal variability of the surface circulation and biological productivity and most importantly, the location and intensity of the Arabian Sea OMZ. A detailed evaluation of the control run is provided in Lachkar et al. (2016).

3. Results

3.1. Contribution of the Arabian Gulf to the Ventilation of the Arabian Sea OMZ

Closing the Gulf considerably increases the intensity of the upper OMZ in the northern Arabian Sea (Figure 1). In particular, the layer between $\sigma_0 = 26$ and $\sigma_0 = 26.8$ that is typically ventilated by the Gulf water (Morrison et al., 1999; Prasad et al., 2001) experiences the largest O₂ decrease. This results in a strong intensification of the OMZ in the upper 200 m, with an increase of the volume of suboxic waters by more than 40% in the AG^{Dry-Gulf} relative to AG^{Control} simulation (Figure 1). In contrast, the volume of suboxia below 200 m increases by a modest 3% in the dry Gulf case in comparison to the control (Figure 1). These



Figure 3. Simulated salinity (psu) in (a) the control run and (b–e) under different warming scenarios along the isopycnal surface $\sigma_0 = 26.4$. (f) Simulated salinity along the same density surface in the closed Gulf run (AG^{Dry-Gulf}).

 O_2 changes are associated with no major change in the density structure or circulation between the two simulations (Figure 1 and Figures S1 and S2 in the supporting information [SI]). Yet, important differences in the salinity of water exist between the two runs. Indeed, the closure of the Gulf causes an important decrease of salinity across the upper ocean, a consequence of the halted ventilation of the upper OMZ by the high-salinity Gulf waters (Figure 1). We hence conclude that the Gulf outflow strongly modulates the intensity of the upper OMZ in the northern Arabian. Next, we explore how the warming of the Gulf waters may affect the ventilation of the Arabian Sea OMZ.

3.2. Effects of the Arabian Gulf Warming on the Ventilation of the OMZ

The simulated warming of the Gulf leads to an important expansion of suboxia in the upper OMZ in the northern Arabian Sea (Figure 1). Under strong warming scenarios (i.e., AG^{4C} and AG^{2C^*}), the intensification of the OMZ is nearly as strong as in the case of the dry Gulf (Figure 1). For instance, the suboxic volume increases in the upper 200 m by 5% under 1 °C warming and by up to 20% and around 45% in the AG^{4C} and AG^{2C^*} warm Gulf runs, respectively (Figure 1e).

In order to better understand the drivers of these changes, we examine the water density across the Strait of Hormuz (56.2°E) that connects the Gulf to the Arabian Sea (Figure 2). We find that the warming of the Gulf waters leads to a decrease in the potential density of the Gulf outflow at the bottom of the Strait of Hormuz (60–80 m) from 26.5 ± 0.1 in the nonperturbed simulation to 25.7 ± 0.1 under the 4 °C warming scenario (Figure 2). Consequently, the more buoyant Gulf outflow waters are not as effective at ventilating the upper (150–250 m) OMZ because they spread at depths which are shallower (100–150 m; Figure 2e and Figure S3 in the SI). The reduction in the ventilation of intermediate layers is also illustrated by the distribution of salinity along the $\sigma_0 = 26.4$ surface depicted in Figure 3. Indeed, the intrusion of the saline Gulf water into the Gulf of Oman decreases as the warming anomaly intensifies. Under the 4 °C Gulf warming (AG^{4C}), the reduction of high salinity in the northern Arabian Sea is nearly as strong as in the case of a dry Gulf.

It is worth noting that the suboxic volume expands more in the AG^{2C^*} relative to the AG^{4C} despite a larger reduction of the Gulf outflow in the latter. This is because the warming is applied over the whole model





Figure 4. Relative change (%) of (a) domain-integrated denitrification in the top 200 m (solid) and below 200 m (dashed) and (b) net primary production (NPP; green) and export production at 200 m (purple) north of 20°N under different perturbation simulations relative to the control run.

domain in the AG^{2C^*} run, causing a domain-wide increase in stratification and decrease in dissolved oxygen in the upper 100 m (Figures S4 and S7 in the SI). This adds up to the effect of reduced ventilation by Gulf waters, leading to a stronger OMZ response. We therefore conclude that the concurrent warming expected over the Arabian Sea is likely to enhance rather than weaken the OMZ response to the warming of the Gulf.

Interestingly, the increase in the suboxic volume becomes weaker under the stronger warming anomalies below 200 m (Figure 1e). In particular, the suboxic volume increases by less than 10% in the AG^{4C} and AG^{2C*} strong warming runs but by around 20% in the AG^{2C} milder warming run. This suggests the existence of a nonlinear response of suboxia to increasing warming beyond which additional warming causes a reduced OMZ response, particularly at depth. In order to understand this nonlinear OMZ response to the Gulf warming, we next consider the effects of denitrification changes on oxygen concentrations at depth.

3.3. Biogeochemical Feedbacks of Reduced OMZ Ventilation

The intensification of the Arabian Sea OMZ results in an important increase in denitrification near the surface. For instance, denitrification is enhanced by 15% and nearly 50% in the upper 200 m in the AG4C and AG^{2C*} strong warming runs (Figure 4a). Below 200 m, denitrification increases by 18% when the Gulf is warmed by 2 °C but only 3% in response to a 4 °C warming perturbation. Denitrification even slightly decreases by around 3% at depth under the AG^{2C*} warming scenario or in the case of a dry Gulf. The enhanced denitrification depletes the subsurface nitrate pool (Figures S5, S6, and S8 in the SI). As nitrification is shut down under suboxic conditions, expanding suboxic volume causes a reduction in nitrification that contributes to nitrate depletion too. However, only enhanced denitrification reduces the total inventory of inorganic nitrogen $(NO_2^- + NH_4^+)$ and hence increases nutrient limitation. The depletion of subsurface nitrate causes a reduction of net primary production (NPP) as well as of export fluxes (Figure 4b). This NPP reduction is largest under strong warming scenarios. For instance, under the AG^{2C*} warming scenario, NPP, and export production at 200 m decrease by up to 13% and 11% north of 20°N, respectively. The negative anomalies in NO₂ concentrations caused by enhanced denitrification are redistributed horizontally,

as circulation acts to reduce the amplitude of the nitrate depletion in the northern Arabian Sea through mixing with surrounding NO_3^- -richer waters, thus spreading the nitrate reduction beyond the northwest Arabian Sea where denitrification is active (Figures S6 and S8 in the SI). This leads to the reduction in productivity and export fluxes to affect the whole Arabian Sea region. Indeed, the domain (excluding the Gulf) integrated NPP and export fluxes decrease by 4% to 6% in the AG^{2C^*} and AG^{4C} runs. A set of additional sensitivity simulations where denitrification was turned off confirms that most of this reduction is driven by enhanced denitrification. Indeed, NPP and export fluxes show a much weaker response to Gulf warming in the absence of denitrification relative to the control (Figure S9 in the SI). This reduction in productivity and export fluxes the biological consumption of oxygen at depth and hence contributes to limit the OMZ intensification below 200 m obtained under strong warming scenarios.

3.4. Role of Salinity Changes

Recent observations suggest an important increase in the surface salinity in several locations of the Gulf (e.g., Al Yamani et al., 2017). For instance, it has been reported that the salinity has increased in the Kuwait Bay over the period 1982–2015 because of a concomitant reduction in Chatt Al Arab discharge in the northern Arabian Gulf (Al Yamani et al., 2017). In general, most future model projections show a salinity increase in the Gulf as a consequence of increased evaporation under climate change (Levang & Schmitt, 2015). For instance, CMIP5 multimodel average indicates an increase in surface salinity by around 0.5 practical

salinity unit (psu) under the highest emission scenario (RCP8.5) by 2100 (Levang & Schmitt, 2015). Yet, these projections are associated with important uncertainties caused by (i) the current poor understanding of the future changes in local air moisture especially in wintertime where the Gulf outflow is densest and most effective in ventilating the Arabian Sea intermediate waters and (ii) the strong sensitivity of Gulf salinity to the inflow of Indian Ocean Surface Water, highly vulnerable to monsoon wind changes (Bower et al., 2000; Kämpf & Sadrinasab, 2006).

In order to explore how changes in salinity of the Gulf water may affect the OMZ response to Gulf warming, we performed two additional sensitivity simulations at a slightly coarser resolution of $(1/12)^{\circ}$ where surface salinity was increased in the Gulf by 0.5 and 1 psu, respectively. These simulations were compared to a new control run performed at an identical resolution $(1/12)^{\circ}$. We find that an increase of the Gulf surface salinity by 0.5 psu that is concomitant with a warming of 4 °C, consistent with CMIP5 model projections, still leads to an intensification of the OMZ that is similar to the simulation with no change in salinity. Yet, the denitrification increase is slightly reduced in comparison to the simulation with no change in salinity. Even a stronger Gulf salinity increase by 1 psu is still associated with an important increase of both denitrification and the suboxic volume. Yet, the expansion of suboxia and the intensification of denitrification are nearly 50% weaker relative to the warming scenario assuming no change in surface salinity (Figures S10 and S11 in the SI).

4. Discussion

4.1. Sensitivity to Winter Warming

To further understand the sensitivity of the OMZ ventilation to the warming of the Gulf, we carried out three additional sensitivity simulations that consisted in (1) two simulations where we applied, respectively, a 2 °C and a 4 °C warming to the Gulf throughout the year (similar to simulations AG^{2C} and AG^{4C}) and (2) a simulation where the 2 °C warming was imposed only during winter months (December through February). A comparison of the O₂ and salinity distributions along the isopycnal surface $\sigma_0 = 26.5$ in the three simulations suggests that the warming during winter months is responsible for most of the OMZ intensification (Figures S10 and S11 in the SI). This is consistent with the fact that the Gulf outflow is substantially stronger and denser during wintertime (Bower et al., 2000).

4.2. Sensitivity to Red Sea Warming

Previous studies suggest that the Red Sea (RS) does contribute to the ventilation of the Arabian Sea (Olson et al., 1993). Similarly to the Gulf, the RS is a semienclosed sea that is vulnerable to amplified warming (Chaidez et al., 2017). To explore the sensitivity of the Arabian Sea OMZ to ongoing and future warming of the RS, we carried out two additional simulations where we applied a 2°C and 4°C warming over the RS while keeping the surface temperature elsewhere unchanged. Furthermore, we made a third simulation where the RS was closed. All these three simulations were made at the highest resolution $(1/24)^{\circ}$ in order to represent the sea floor geometry at the narrow Strait of Bab el Mendeb as realistically as possible. The warming of the RS is shown to reduce the ventilation of the OMZ in the northern Arabian Sea, thus leading to its intensification. But unlike with the Gulf warming, the OMZ intensifies essentially at depth (>400 m) in response to the warming of the RS. In contrast, the response of the upper OMZ (<200 m) is much weaker (Figure S12 in the SI). Additionally, the response of the OMZ scales more linearly with the size of the warming perturbation and only becomes large (>20%) under the 4 °C strong warming scenario (Figure S12 in the SI). The higher sensitivity of the deeper OMZ to the RS warming can be explained by the higher density of the RS water $\sigma_0 = 27-27.4$ that ventilates deeper layers in the northern Arabian Sea (600–1,000 m; Beal et al., 2000; Olson et al., 1993). The more linear response of the OMZ to the RS warming can be linked to the relatively weak changes in the upper ocean suboxia, thus limiting the effect of denitrification feedback on subsurface nitrogen budgets and productivity and hence O₂ consumption at depth (Figure S12 in the SI).

4.3. Implications and Caveats

The Gulf and the RS are separated from the Arabian Sea by narrow and shallow sills (\approx 80 m at the Strait of Hormuz and \approx 130 m at Bab El Mandeb). Therefore, during glacial periods the Arabian Gulf was essentially dry and the Red Sea outflow was strongly reduced (e.g., by 85% during the last glacial maximum) because of the glacial low sea level (Rohling & Zachariasse, 1996; Siddall et al., 2003). Our study suggests that the separation of the Arabian Sea from its two marginal seas may have led to a reduction of the ventilation of the

intermediate layers and hence an intensification of the Arabian Sea OMZ during glacial periods. Pichevin et al. (2007) proposed that the cutoff of the two marginal seas outflows during glacial periods may have led to a reorganization of the large-scale circulation in such a way that the ventilation of intermediate layers in the northern Indian Ocean by the Antarctic Intermediate Water from the south is further enhanced. These authors further suggested that this could have led to an increase in the ventilation and a weakening of the Arabian Sea OMZ. Our study does not support this hypothesis as we did not find any noticeable change in the large-scale circulation following the closure of the Gulf (or of the Red Sea). In contrast, our results are consistent with the findings of McCreary et al. (2013) that show an important contribution of the Gulf outflow to the ventilation the OMZ in the Sea of Oman. These authors also found that the ventilation of the OMZ by the Gulf outflow may contribute to the observed eastward shift of the Arabian Sea OMZ (from the region of maximum productivity in the western Arabian Sea).

Our study highlights the importance of the semienclosed seas for open ocean large-scale biogeochemistry. The Arabian Sea marginal seas are typically poorly represented in current generation coarse-resolution climate models as is the Arabian Sea OMZ (Bopp et al., 2013; Cocco et al., 2013). These biases have been linked to several factors including the poor representation of mesoscale eddies and mixing and biases in the representation of biogeochemical processes (Gnanadesikan et al., 2013; Lachkar et al., 2016; Oschlies et al., 2018). Here, we propose that the poor representation of the Arabian marginal seas in these models can also contribute to those biases.

Finally, our study has several caveats and limitations. Among the study's main limitations is the lack of a representation of nitrogen fixation in the model. This could artificially amplify the effect of denitrification on the nitrogen budget and ultimately lead to an overestimated feedback of suboxia on productivity and O_2 consumption. Yet, we believe this effect to likely remain small as denitrification is thought to dominate over N2 fixation in the Arabian Sea (Bange et al., 2005). Furthermore, although observations suggest a tight coupling between denitrification and N_2 fixation on timescales of thousands of years (Gruber, 2008), the persistent excess phosphate over nitrate in the Arabian Sea indicates a weak coupling between the two processes regionally and on shorter timescales (Gruber and Sarmiento 1997). This is also supported by evidence from paleoclimate records (e.g., Altabet et al., 1999, Altabet et al., 2002). Another major caveat is the use of idealized perturbations that assume that SST changes are uniform either in space or time and hence do not take into account the full complexity of realistic future trajectories. Additionally, the effect of the Gulf warming is taken in isolation of other potential concomitant environmental changes such as changes in large-scale winds (e.g., Lachkar et al., 2018) and circulation or local perturbations such as pollution and eutrophication. Combined, these stressors can lead to antagonistic effects that dampen the overall response or synergistic effects that cause a larger response than that be predicted from individual perturbations considered separately. For instance, recent observations suggest an important decline in the Gulf oxygen content due to pollution and eutrophication (Al Said et al., 2018). This could enhance the effect of warming and lead to stronger OMZ intensification.

5. Summary and Conclusions

A set of high-resolution model simulations reveals a strong vulnerability of the Arabian Sea OMZ to the warming of the Gulf. As they exit the Strait of Hormuz, the warmer Gulf waters gain buoyancy and spread at a shallower depth in the northern Arabian Sea. This results in a decrease of the ventilation of the intermediate (200-300 m) depths, thus causing an expansion of the suboxic volume and an intensification of the upper OMZ. The increase of denitrification that results from the expanded suboxia depletes subsurface nitrate concentrations and leads to a reduced productivity and O₂ consumption at depth. Under moderate warming, this partially compensates the effect of reduced ventilation on oxygen. Under strong warming scenarios, the denitrification feedback can be strong enough to offset the effect of reduced ventilation at depth. The projected future increase in Gulf water salinity can partially mitigate the effect of Gulf warming only to a limited extent as the impact of warming is likely to dominate. Our study suggests that the observed fast warming of the Gulf over the last few decades may have contributed to the recent expansion of the OMZ in the northern Arabian Sea. The projected future strong warming of the Gulf may further enhance this tendency and lead to an increase in the frequency and severity of hypoxic and anoxic events in the northern Arabian Sea together with a strengthening of denitrification, with a potential for a profound alteration of the ecosystems and the biogeochemistry of the Arabian Sea and the Indian Ocean at large.



Acknowledgments

Support for this research has come from the Center for Prototype Climate Modeling (CPCM), the New York University Abu Dhabi (NYUAD) Research Institute. Computations were performed at the High Performance cluster (HPC) of NYUAD, Dalma. We thank B. Marchand and M. Barwani from the NYUAD HPC team for technical support. The authors declare that they have no competing financial interests. The data used for forcing and validating the model are publicly available online and can be accessed from cited references. The model code can be accessed online (http://www. romsagrif.org). The model outputs are available online (https://zenodo.org/ record/2065158).

References

- Al Rashidi, T. B., El-Gamily, H. I., Amos, C. L., & Rakha, K. A. (2009). Sea surface temperature trends in Kuwait Bay, Arabian Gulf. Natural Hazards, 50(1), 73–82.
- Al Said, T., Naqvi, S., Al-Yamani, F., Goncharov, A., & Fernandes, L. (2018). High total organic carbon in surface waters of the northern Arabian Gulf: Implications for the oxygen minimum zone of the Arabian Sea. *Marine Pollution Bulletin*, 129(1), 35–42.
- Al Yamani, F., Yamamoto, T., Al-Said, T., & Alghunaim, A. (2017). Dynamic hydrographic variations in northwestern Arabian Gulf over the past three decades: Temporal shifts and trends derived from long-term monitoring data. *Marine Pollution Bulletin*, 122(1-2), 488–499. Bange, H. W., Naqvi, S. W. A, & Codispoti, L. (2005). The nitrogen cycle in the Arabian Sea. *Progress in Oceanography*, 65(2-4), 145–158.
- Barnier, B., Siefridt, L., & Marchesiello, P. (1995). Thermal forcing for a global ocean circulation model using a three-year climatology of ecmwf analyses. *Journal of Marine Systems*, 6(4), 363–380.

- Bopp, L., Resplandy, L., Orr, J. C., Doney, S. C., Dunne, J. P., Gehlen, M., et al. (2013). Multiple stressors of ocean ecosystems in the 21st century: Projections with CMIP5 models. *Biogeosciences*, 10, 6225–6245.
- Bower, A. S., Hunt, H. D., & Price, J. F. (2000). Character and dynamics of the Red Sea and Persian Gulf outflows. Journal of Geophysical Research, 105(C3), 6387–6414.
- Bower, A. S., Johns, W. E., Fratantoni, D. M., & Peters, H. (2005). Equilibration and circulation of Red Sea outflow water in the Western Gulf of Aden. Journal of Physical Oceanography, 35(11), 1963–1985.
- Chaidez, V., Dreano, D., Agusti, S., Duarte, C. M., & Hoteit, I. (2017). Decadal trends in Red Sea maximum surface temperature. Scientific Reports, 7(1), 8144.
- Cocco, V., Joos, F., Steinacher, M., Frölicher, T., Bopp, L., Dunne, J., et al. (2013). Oxygen and indicators of stress for marine life in multi-model global warming projections. *Biogeosciences*, 10(3), 1849–1868.
- Codispoti, L., Brandes, J. A., Christensen, J., Devol, A., Naqvi, S., Paerl, H. W., & Yoshinari, T. (2001). The oceanic fixed nitrogen and nitrous oxide budgets: Moving targets as we enter the anthropocene? *Scientia Marina*, 65(S2), 85–105.
- da Silva, A. M., Young, C. C., & Levitus, S. (1994). Atlas of surface marine data 1994, vol. 4: Anomalies of fresh water fluxes. NOAA Atlas, NESDIS 9.
- Gnanadesikan, A., Bianchi, D., & Pradal, M.-A. (2013). Critical role for mesoscale eddy diffusion in supplying oxygen to hypoxic ocean waters. *Geophysical Research Letters*, 40, 5194–5198. https://doi.org/10.1002/grl.50998
- Gomes, H., do Rosário, J. I., Goes, S., Matondkar, E. J., Buskey, S., Basu, S. P., & Thoppil, P. (2014). Massive outbreaks of noctiluca scintillans blooms in the Arabia Sea due to spread of hypoxia. *Nature Communications*, *5*, 4862.
- Gruber, N., Frenzel, H., Doney, S. C., Marchesiello, P., McWilliams, J. C., Moisan, J. R., et al. (2006). Eddy-resolving simulation of plankton ecosystem dynamics in the California Current System. *Deep Sea Research Part I: Oceanographic Research Papers*, 53(9), 1483–1516.
- Hoegh-Guldberg, O., Cai, R., Poloczanska, E. S., Brewer, P. G., Sundby, S., Hilmi, K., et al. (2014). The Ocean. In V. R. Barros, C. B. Field, D. J. Dokken, M. D. Mastrandrea, K. J. Mach, T. E. Bilir, M. Chatterjee, K. L. Ebi, Y. O. Estrada, R. C. Genova, B. Girma, E. S. Kissel, A. N. Levy, S. MacCracken, P. R. Mastrandrea, & L. L. White (Eds.), *Climate Change 2014: Impacts, Adaptation, and Vulnerability. Part B: RegionalAspects. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panelon Climate Change* (pp. 1655–1731). Cambridge, UK and New York: Cambridge University Press.
- Ito, T., Minobe, S., Long, M. C., & Deutsch, C. (2017). Upper ocean O₂ trends: 1958–2015. Geophysical Research Letters, 44, 4214–4223. https://doi.org/10.1002/2017GL073613
- Kämpf, J., & Sadrinasab, M. (2006). The circulation of the Persian Gulf: A numerical study. Ocean Science, 2(1), 27-41.
- Lachkar, Z., Lévy, M., & Smith, S. (2018). Intensification and deepening of the Arabian Sea oxygen minimum zone in response to increase in Indian monsoon wind intensity. *Biogeosciences*, 15(1), 159–186.
- Lachkar, Z., Smith, S., Lévy, M., & Pauluis, O. (2016). Eddies reduce denitrification and compress habitats in the Arabian Sea. Geophysical Research Letters, 43, 9148–9156. https://doi.org/10.1002/2016GL069876
- Large, W. G., McWilliams, J. C., & Doney, S. C. (1994). Oceanic vertical mixing: A review and a model with a nonlocal boundary layer parameterization. *Reviews of Geophysics*, 32(4), 363–403.
- Levang, S. J., & Schmitt, R. W. (2015). Centennial changes of the global water cycle in CMIP5 models. *Journal of Climate*, *28*(16), 6489–6502. Marchesiello, P., Debreu, L., & Couvelard, X. (2009). Spurious diapycnal mixing in terrain-following coordinate models: The problem and a solution. *Ocean Modelling*, *26*(3-4), 156–169.
- McCreary, J. P. J., Yu, Z., Hood, R. R., Vinaychandran, P., Furue, R., Ishida, A., & Richards, K. J. (2013). Dynamics of the Indian Ocean oxygen minimum zones. *Progress in Oceanography*, 112, 15–37.
- Middelburg, J. J., Soetaert, K., Herman, P. M., & Heip, C. H. (1996). Denitrification in marine sediments: A model study. *Global Biogeochemical Cycles*, 10(4), 661–673.
- Morrison, J. M., Codispoti, L. A., Smith, S. L., Wishner, K., Flagg, C., Gardner, W. D., et al. (1999). The oxygen minimum zone in the Arabian Sea during 1995. Deep Sea Research Part II: Topical Studies in Oceanography, 46(8-9), 1903–1931.
- Olson, D. B., Hitchcock, G. L., Fine, R. A., & Warren, B. A. (1993). Maintenance of the low-oxygen layer in the central Arabian Sea. *Deep Sea Research Part II: Topical Studies in Oceanography*, 40(3), 673–685.
- Oschlies, A., Brandt, P., Stramma, L., & Schmidtko, S. (2018). Drivers and mechanisms of ocean deoxygenation. *Nature Geoscience*, 11, 467–473.
- Pichevin, L., Bard, E., Martinez, P., & Billy, I. (2007). Evidence of ventilation changes in the Arabian Sea during the Late Quaternary: Implication for denitrification and nitrous oxide emission. *Global Biogeochemical Cycles*, 21, GB4008. https://doi.org/10.1029/ 2006GB002852
- Piontkovski, S., & Al-Oufi, H. (2015). The Omani shelf hypoxia and the warming Arabian Sea. International Journal of Environmental Studies, 72(2), 256–264.
- Prasad, T., Ikeda, M., & Kumar, S. P. (2001). Seasonal spreading of the Persian Gulf water mass in the Arabian Sea. Journal of Geophysical Research, 106(C8), 17,059–17,071.
- Queste, B. Y., Vic, C., Heywood, K. J., & Piontkovski, S. A. (2018). Physical controls on oxygen distribution and denitrification potential in the north west Arabian Sea. *Geophysical Research Letters*, 45, 4143–4152. https://doi.org/10.1029/2017GL076666
- Resplandy, L., Lévy, M., Bopp, L., Echevin, V., Pous, S., Sarma, V., & Kumar, D. (2012). Controlling factors of the oxygen balance in the Arabian Sea's OMZ. *Biogeosciences*, 9(12), 5095–5109.
- Risien, C. M., & Chelton, D. B. (2008). A global climatology of surface wind and wind stress fields from eight years of quikscat scatterometer data. *Journal of Physical Oceanography*, *38*(11), 2379–2413.

Beal, L. M., Ffield, A., & Gordon, A. L. (2000). Spreading of red sea overflow waters in the Indian Ocean. Journal of Geophysical Research, 105(C4), 8549–8564.

Rohling, E., & Zachariasse, W. (1996). Red sea outflow during the last glacial maximum. Quaternary International, 31, 77-83.

- Shchepetkin, A. F., & McWilliams, J. C. (2005). The regional oceanic modeling system (ROMS): A split-explicit, free-surface, topography-following-coordinate oceanic model. *Ocean Modelling*, 9(4), 347–404.
- Siddall, M., Rohling, E. J., Almogi-Labin, A., Hemleben, C., Meischner, D., Schmelzer, I., & Smeed, D. (2003). Sea-level fluctuations during the last glacial cycle. *Nature*, 423(6942), 853.

Smith, W. H., & Sandwell, D. T. (1997). Global sea floor topography from satellite altimetry and ship depth soundings. Science, 277(5334), 1956–1962.

Strong, A. E., Liu, G., Skirving, W., & Eakin, C. M. (2011). NOAA's coral reef watch program from satellite observations. *Annals of GIS*, 17(2), 83–92.

You, Y. (1998). Intermediate water circulation and ventilation of the Indian Ocean derived from water-mass contributions. Journal of Marine Research, 56(5), 1029–1067.